Author’s response to reviews

Title: Incidence of Viral Respiratory Infections among Outpatient and Hospitalized Children Aged [less than or equal to] 5 Years and Its Associated Cost in Buenos Aires, Argentina

Authors:

- Debora Marcone (debroamarcone@hotmail.com)
- Lizette O Durand (vcn6@cdc.gov)
- Eduardo Azziz-Baumgartner (eha9@cdc.gov)
- Santiago Vidaurreta (svidaurreta@gmail.com)
- Jorge Ekstrom (jorgeekstrom@hotmail.com)
- Guadalupe Carballal (gcarballal@cemic.edu.ar)
- Marcela Echavarria (Mechavarria@cemic.edu.ar)

Version: 6 Date: 21 September 2015

Author’s response to reviews: see over
Dear Editor-in-Chief,

Thank you for working with us to further polish the manuscript titled “Incidence of Viral Respiratory Infections among Outpatient and Hospitalized Children Aged ≤5 Years and Its Associated Cost in Buenos Aires, Argentina.”

I agree with your recommendations and have strived to incorporate all into the revised manuscript. The manuscript now has the following edits:

1. We have included the STROBE checklist. When necessary, we elaborated on the answers in the space provided.
2. As recommended, all main section headings (Background, Methods, Results etc are in line with the journal Author Instructions.
3. The source of funding has been elaborating on in the disclaimer section, which has been moved and placed at the end of the Acknowledgement section.
4. All authors have read and approved the final version of the manuscript.
5. Figure titles and legends have been placed inside of the main document.

If you have any further recommendations to strengthen our manuscript, please feel free to contact us. Thank you again.

Eduardo Azziz-Baumgartner, MD, MPH
Influenza Division, U.S. CDC
1600 Clifton Rd, NE, MS A32
Atlanta, GA 30329-4018